Author's response to reviews

Title: Far from easy and accurate - detection of metabolic syndrome by general practitioners

Authors:

Eeva-Eerika Helminen (eeva-eerika.helminen@fimnet.fi)
Pekka Mäntyselkä (pekka.mantyselka@uku.fi)
Irma Nykänen (irma.nykanen@uku.fi)
Esko Kumpusalo (esko.kumpusalo@uku.fi)

Version: 5 Date: 20 November 2009

Author's response to reviews:

We only removed extra double spaces (in two places) and changed the typography of column labels (from Italic to Bold) in Table 1 and 2. Otherwise it is allright and we accept it now as it is.